Author's response to reviews

Title: Using Intervention Mapping to develop a programme to prevent sexually transmittable infections, including HIV, among heterosexual migrant men.

Authors:

Mireille E.G. Wolfers (wolfersm@ggd.rotterdam.nl)
Caty van den Hoek (vandenhoekk@ggd.rotterdam.nl)
Johannes Brug (j.brug@erasusmc.nl)
Onno de Zwart (dezwarto@ggd.rotterdam.nl)

Version: 7 Date: 2 July 2007

Author's response to reviews:

1) Abstract - Conclusion is changed into Conclusions
2) Title page - text 'authors affiliations' removed
3) Website - pg 8 is removed from the text, and cited from the main reference list now.
4) Figures 1 is cropped now